

REPLY TO A REVIEWER AND AN EDITOR



March 29, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 16623-edited.docx).

Title: Changing common sense: anti-platelet/coagulation therapy against cirrhosis

Author: Yoshihiro Ikura, Tatsuya Osuga

Name of Journal: *World Journal of Hepatology*

ESPS Manuscript NO: 16623

The manuscript has been improved according to the suggestions of a reviewer and an editor:

1) COI statement forms are attached.

2) Audio Core Tip was added.

Thank you again for publishing our manuscript in the *World Journal of Hepatology*.

Sincerely yours,

A handwritten signature in black ink that reads 'Yoshihiro Ikura'.

Yoshihiro Ikura, MD

Department of Pathology

Takatsuki General Hospital

1-3-13, Kosobecho

Takatsuki 569-1192,

Japan

E-mail: ikura@ajk.takatsuki-hp.or.jp

Phone & Fax: 072 684 0295